Author's response to reviews

Title: Elastography in predicting preterm delivery in asymptomatic, low-risk women: a prospective observational study.

Authors:

Slawomir WOZNIAK (slawwo7572@gmail.com)
Piotr CZUCZWAR (czuczwar@wp.pl)
Piotr SZKODZIAK (piotr.szkodziak@gmail.com)
Pawel MILART (pmilart@gmail.com)
Ewa WOZNIAKOWSKA (ewa.wozniakowska@gmail.com)
Tomasz PASZKOWSKI (tompasz@tlen.pl)

Version: 3
Date: 18 April 2014

Author's response to reviews:

Dear editorial board of BMC Pregnancy and Childbirth.

Please find enclosed the revised manuscript: "Elastography in predicting preterm delivery in asymptomatic, low-risk women." by Slawomir Wozniak et al. We followed all the suggestions and made all the requested corrections:

1. An acknowledgements section was added between the authors’ contribution and reference list.
2. An authors’ contribution section was added before the acknowledgements.
3. Competing interests section after the Conclusions was added.
4. Figure legend header was added.
5. The tables were removed from the additional files and placed in the main manuscript.

We hope that the editorial board will agree on the interest of this study.

Sincerely yours,
Piotr Czuczwar and Slawomir Wozniak on behalf of the authors.

Corresponding author: Piotr Czuczwar at Medical University of Lublin, 3rd Chair and Department of Gynecology, 20-954 Lublin, ul. Jaczewskiego 8, Poland, czuczwar@wp.pl, phone number: +48817244848